

Received 22 August 2023, accepted 30 September 2023, date of publication 12 October 2023, date of current version 18 October 2023. Digital Object Identifier 10.1109/ACCESS.2023.3324292

RESEARCH ARTICLE

Grid Deployment Scheme for Enhancing Network Performance in Underwater Acoustic Sensor Networks

AHMED ALJUGHAIMAN^(D), (Member, IEEE) Department of Computer Networks and Communications, College of Computer Sciences and Information Technology, King Faisal University, Al-Ahsa 31982, Saudi Arabia

e-mail: aaaljughaiman@kfu.edu.sa

This work was supported by the Deputyship for Research and Innovation, Ministry of Education, Saudi Arabia, under Project INST187.

ABSTRACT Researchers have achieved significant progress on the problem of Sensor Node Deployment (SND) in Underwater Acoustic Sensor Networks (UASNs) in terms of enhancing underwater communication. However, most studies focus on enhancing coverage, connectivity, selective network performance metrics, and/or deployment expenses at the cost of other critical factors. Given the limited resources and difficulties with using renewable energy sources to recharge UASN batteries, it is necessary to minimize Energy Consumption (EC) through the node deployment mechanism to extend network lifetime. Therefore, in this paper, the author proposes the Distributed Deployment Optimization algorithm using Grid-based Depth Adjustable (DDOGDA) based on grid node deployment for 3D network architecture. With this model, this paper endeavors to monitor the underwater environment with the minimum number of underwater nodes while meeting the QoS requirements of a tsunami-monitoring application in the Solomon Islands. The proposed algorithm considers Geographic Information System (GIS) data, non-environmental factors, and the unique characteristics of underwater sensors. These factors were used to provide guidance on how to place nodes properly to achieve certain objectives. Herein, the proposed algorithm is compared to six other node deployment algorithms. Simulation results indicate that our proposed algorithm surpasses random, tetrahedron, cuboid, triangular, pipeline, and grid node deployment in terms of End-to-End Delay (E2ED), EC, and Packet Delivery Ratio (PDR) by 266%, 183%, and 22%, respectively.

INDEX TERMS Disaster-monitoring application, end-to-end delay, energy consumption, network performance, node deployment, packet delivery ratio, time-sensitive application, underwater acoustic sensor networks.

I. INTRODUCTION

With the recent advances in UASNs, underwater nodes are a promising technology that can support several types of underwater applications, such as those for environmental monitoring, assisted navigation, disaster prevention, sports, and the military [1], [2], [3], [4], [5]. Real-time monitoring in UASNs is very important to time-sensitive monitoring applications as it is necessary to avert or warn of potential disasters (e.g., flood, volcanic eruption, tsunami, earthquake,

The associate editor coordinating the review of this manuscript and approving it for publication was Jiajia Jiang^D.

and pipeline malfunction/break/leak) which can have severe impact on human and marine life as well as national, regional, and global economies [6]. Therefore, a new and reliable mechanism is necessary to achieve the requirements of target applications through an advanced UASN [7]. Different underwater applications have different requirements and hence different node deployments can be used to meet those requirements [8]. In comparison to Terrestrial Wireless Sensor Network (TWSN) communication, each underwater application must consider the details of the target area and the requirements of the target underwater application to properly handle high and variable propagation delay,

limited bandwidth, high transmission loss, multipath effect, and Doppler spread [9], [10], [11], [12]. In addition, nonenvironmental and environmental factors, such as network size, packet size, network load, current, seismic activity, and noise must be considered as these factors make designing efficient underwater communication more challenging. Network topology also has a strong impact on network performance [13]. All these factors must be taken into consideration when designing an optimal algorithm for SND in order to make the model an efficient approach for underwater applications.

In the last decade, node deployment algorithms for TWSNs have been investigated, but there are significant differences between TWSNs and UASNs, so this research is not directly applicable to UASNs [14]. For example, in TWSNs, node deployment is applied to the 2D deployment space, while applications in UASNs require deployment in the 3D deployment space [15]. While 2D is sometimes used in UASNs, it limits the functionality of the network because the 3D nature of the environment cannot be considered. This is why at least 3D is preferred for underwater applications. Node deployment in 2D for TWSNs has been studied intensively, but the solutions obtained cannot be implemented in UASNs, as most underwater applications rely on 3D [16], [17], [18]. In comparison to TWSNs, communication in UASNs is not stable, there is the need to address the unique characteristics of acoustic channel, and there is limited battery coupled with the inability to recharge using any sort of renewable energy [19]. Furthermore, the underwater environment is an extremely severe one in comparison with that of TWSNs, which correlates to variation of delay, current, depths, and the other characteristics of acoustic channel [20], [21]. These characteristics pose great challenges to the design of an effective mechanism for network deployment in UASNs. Most studies on 2D node deployment for TWSNs also do not consider the topographic information of the target application [15]. This leads to inefficient node deployment mechanisms and poor performance that may not meet the target application's QoS requirements. Therefore, existing node deployment solutions in TWSNs cannot be directly implemented in UASNs and hence a new node deployment mechanism is necessary [22], [23].

Node deployment in UASNs can be categorized into 2D, 3D, and 4D network architectures. In all network architectures, underwater nodes must collaborate to deliver packets to the sink node [24]. These network architectures share similar challenges, such as determining the minimum number of nodes to monitor an Area of Interest (AoI); determining the optimal location of sensor nodes, relay nodes (also known as gateway nodes), and sink nodes (also known as surface stations); determining the optimal depth to achieve full coverage and connectivity; enabling collaboration between nodes to transmit data through multihop paths; and optimizing network performance while minimizing the utilization of the low network resources [25], [26], [27], [28], [29]. To address the above challenges and



FIGURE 1. Components of 3D network architecture.

meet the QoS requirements of target applications, a new algorithm is necessary for UASNs.

A typical underwater network consists of a set number of underwater nodes with limited resources to monitor an AoI [15], [30]. Particularly, a UASN in 3D network architecture consists of sensor nodes, relay nodes, and sink nodes as shown in Figure 1. Optimization of node deployment can be achieved through deploying sensor, relay, and sink nodes appropriately [31]. The location of each one of these nodes plays an important role in meeting the QoS requirements of the target application. When optimizing underwater communication, there are several factors to keep in mind (e.g., coverage, connectivity, sensor deployment, gateway deployment, sink deployment), which have great influence on the performance of the UASN and thus must be duly considered [15], [30]. Designing an efficient node deployment algorithm can enhance network performance. Therefore, the proposed node deployment algorithm considers the coverage area, available data rate, network size, packet size, network load, and other characteristics to achieve better network performance while meeting the QoS requirements of the target application.

The balance of this paper is organized as follows. Section II discusses the background of SND algorithms. Section III discusses related research articles and highlights the unique design characteristics examined in each. Section IV explains the proposed SND algorithm. Section V presents the details of the experimental setup and evaluation metrics utilized in this study and compares the results with those of other algorithms. Section VI concludes this paper while Section VII describes some future directions.

II. BACKGROUND

SND has a strong impact not only on coverage and connectivity, but also on the QoS of UASNs [27], [32], [33], [34], [35]. Utilizing a large number of sensors to monitor



FIGURE 3. Underwater nodes.

Random Deterministic Heterogenous Homogenous Static Self-adjustment Movement-assisted

FIGURE 4. Node deployment strategies.

an AoI can address scalability and reliability concerns but comes at an increased total cost [36]. Underwater nodes can reach one another through single-hop single-path, single-hop multi-path, multi-hop single-path, or multi-hop multi-path as shown in Figure 2. Therefore, node deployment planning is crucial to fully meeting the requirements of underwater applications [37].

SND algorithms can be classified into optimal location of the sensor nodes, relay nodes, and sink nodes, as shown in Figure 3. Each one of these nodes can act as a sensor node by collecting data within its sensing range and reporting it to the sink node. The location of each of these nodes plays a crucial role in achieving full coverage and connectivity as well as achieving high performance in underwater communication [6], [38], [39]. Therefore, it is necessary to consider the locations of all these nodes when developing an optimal SND algorithm. The location of all these nodes also depends on the size of the simulation area. Different applications are required to monitor different sizes of AoIs. In a large-size simulation area, a single sink node may not be enough to achieve high network performance, indicating it is necessary to utilize multiple sinks for such applications. In contrast, a small AoI can be monitored properly with a single sink node.

The deployment of underwater nodes in UASNs can be classified into different deployment strategies [24], [28], [36], [37], [40] that can be either random or deterministic, as shown in Figure 4 [41]. Addressing SND issues can be achieved via heterogeneous sensor network or homogeneous sensor network [22], [42], [43]. Moreover, resolving coverage and connectivity issues can be achieved more efficiently through deterministic node deployment as the method can minimize the number of underwater nodes needed [36], [44]. In both random and deterministic node deployments, underwater nodes can be organized in a triangular, grid, or hexagonal pattern. Moreover, underwater nodes can cover an AoI sparsely or densely. When the distance between the underwater nodes is large, resolving some issues-such as full connectivity and detecting node failure-becomes more challenging. For example, full connectivity can be achieved through long transmission range and node failure can be detected when no packets are received at the sink node at regular intervals. To avoid false node failure detection, failure is only determined when the sink nodes do not receive any packets from a particular sensor node after an established threshold value is reached. Only after this set time period has passed is the sensor identified as failing. In contrast, network performance may also be affected when the distance between underwater nodes is short, due to the increased chance of interference between neighboring nodes, which consumes higher energy and shortens network lifetime [23]. Thus, it is wise to use lower numbers of sensors to meet the QoS requirements of target applications and hence lower the total cost [45]. Additionally, deployment strategy relies on two different data reporting systems, event-driven and on-demand, depending on the type of application. Some underwater applications require underwater nodes to monitor an AoI on an event-driven basis while others require ondemand monitoring. The network lifetime in event-driven and on-demand reporting systems is dependent on the conditions of the severe environment and the requirements of the underwater application. For example, Deep-ocean Assessment and Reporting of Tsunami (DART) requires underwater nodes to report the potential of tsunami regularly, which consumes higher energy and hence shortens network lifetime. In contrast, applications that monitor oil/gas pipelines require event-driven reporting, which means fewer packets need to be transmitted to the sink node, which thereby extends network lifetime.

The main objective of optimal node deployment is to monitor an AoI with the minimum number of sensor nodes to achieve maximum coverage and connectivity while meeting the QoS requirements of an underwater application. Node deployment algorithms can be classified as single-objective or multi-objective [43], as shown in Figure 5. In the former, the algorithms proposed in previous studies have focused on only a single issue, such as coverage or connectivity, or only a single performance metric. In the latter, the algorithms proposed in previous studies have focused on multiple issues. Clearly, addressing node deployment problems with multi-objective approach is more appropriate to achieve optimal solutions for target monitoring applications.



FIGURE 5. SND objective.



FIGURE 6. Impact of tsunami events [46].

As observed from the literature review, most research papers have focused on coverage, connectivity, and/or selected performance metrics while some papers ignored some of these issues. Therefore, in this study, the author aims to resolve SND issues by designing an optimal location method for underwater nodes for a tsunami monitoring application. As can be observed from this section, designing an optimal SND algorithm is a very challenging problem as many factors must be considered in order to achieve the QoS requirements of the target application.

Several destructive tsunami events have occurred over the last decade, each of which resulted in the loss of many human lives. Figure 6 shows the tsunami events that occurred during this time period and notes the impact of each of these events in terms of the number of deaths. In addition, this figure classifies the tsunami events using color coding (i.e., white, yellow, orange, red, and purple), where: (i) white represents a tsunami event where the number of deaths was zero or unknown; (ii) yellow represents an event where the number of deaths was between 1 and 50; (iii) orange indicates the number of deaths was between 51 and 100; (iv) red denotes that the number of deaths was between 101 and 1,000; and (v) purple is used to indicate that the number of deaths was 1,001 or more [46]. The total loss of life during these incidents could have been reduced by the deployment of sensor nodes underwater that would have generated an early warning for evacuation [47], [48]. Tsunamis are caused by underwater earthquakes that result in huge waves of water being



FIGURE 7. DART stations [56].

generated that can drown an entire island [49]. The speed of a tsunami depends on the water depth. The greater the depth of the water, the faster the tsunami. In ultra-deep-water, a tsunami's speed can reach up to 500 mph, but it slows down as it approaches coastal areas [49]. Therefore, it is necessary to deploy underwater sensors to monitor underwater pressure to evaluate the potential of tsunami incidents [50], [51]. More particularly, a tsunami can be detected by analyzing the pressure fluctuations to differentiate between normal waves and tsunami waves [19], [52]. To have an effective tsunami warning system, packets must be transmitted to the sink node successfully with minimal delay as the speed and timeliness of the warning will save human lives [19], [53]. To achieve this type of effective system, it is vital to deploy sensor nodes properly to achieve maximum coverage and connectivity, minimum E2ED and EC, while also maximizing PDR.

Over the years, many devastating tsunamis have occurred across the globe resulting in the deaths of many people [54]. There is clearly a critical need for reliable warning systems to save human lives. In the last decade, UASNs have received a great deal of attention in the area of oceanic research, due to the wide range of underwater applications. However, some ocean research still relies on 1D network architecture where it deploys underwater sensors in the ocean and then retrieves them to analyze the collected data [55]. Clearly, this is not an efficient approach to monitor underwater environments. It is also not a method that can support time-sensitive applications designed to predict natural hazards. Therefore, it is necessary to monitor underwater applications with an automated sensor network that can meet the QoS requirements of the target applications [51]. The underwater disaster application is considered an event-driven reporting system, which must be monitored at all times to save human lives. Currently, there are 39 stations around the planet, as shown in Figure 7, implemented in regions where a destructive tsunami has occurred in the past [46], [55]. The deployment of these stations is part of the DART project that is maintained by the National Oceanic and Atmospheric Administration (NOAA). Each one of these stations is composed of underwater sensors on the ocean floor and a sink node at the surface level, meaning these stations monitoring for the potential of a tsunami incident employ 2D network architecture. The main

objective of the DART system is to provide accurate readings of the underwater pressure with minimal delay to generate early warning for tsunami and avoid issuing false alarms for evacuation [45]. In such a critical application, efficient communication between nodes is very important to transmit the pressure status underwater with minimal delay [52].

III. RELATED WORK

The SND problem is critical for underwater applications as it has a great influence on network performance. SND can be organized in 2D, 3D, or 4D network architecture. The type of network architecture depends on the type of applications that must be monitored. Each application has different requirements and may require different network architectures. Thus, some underwater applications require 2D network architecture, the majority require 3D network architecture, and a very few applications require 4D network architecture in UASNs. The capabilities of sensor nodes in underwater Wireless Sensor Networks (WSNs) have limited resources compared to those of other electronic devices. This means that, due to their limited resources, deploying sensor nodes in a severe environment makes it more challenging to meet the QoS requirements of an application.

To develop a suitable SND algorithm for a target application that also enhances network performance, many issues must be addressed, including coverage holes and connectivity issues. Currently, most studies do not address all these issues when developing an SND algorithm, resulting in degradation of network performance in multiple metrics. Therefore, it is very important to establish the optimal location of sensor nodes, relay nodes, and sink nodes to achieve high performance and meet the QoS requirements of an application.

Jiang et al. [57] proposed a Redeployment based on Virtual Forces (RBVF) algorithm where a node adjusts its depth based on virtual forces algorithm. RBVF focuses on maximal coverage area when redeployment is needed. In [57], the authors also considered water flow force and proposed a Virtual Forces Redeployment based on Energy Consuming (VFRBEC) model to minimize EC. Simulation results show that RBVF can achieve higher coverage and requires less moving distance compared to other algorithms. Moreover, simulation results also indicate that VFRBEC can achieve the same coverage compared to RBVF with lower EC. However, despite these benefits, both RBVF and VFRBEC focus on the coverage problem and ignore connectivity, cost, and other performance metrics. In addition, while both RBVF and VFRBEC enhance the coverage area, some areas remained uncovered.

Similar to RBVF [57], Liu et al. [20] proposed a distributed node Deployment Algorithm Based on Virtual Forces (DABVF) to achieve excellent coverage while reducing EC. DABVF algorithm takes into consideration node density, node residual energy, and node mobility to optimize the location of underwater nodes. Simulation results reveal that DAVBF achieves higher network coverage, reduces EC, balances node residual energy, and enhances the location of node distribution compared to other node deployment algorithms. DABVF focuses on resolving coverage holes by achieving k-coverage, which involves having several underwater nodes monitor an identical area. This indicates that in this scenario numbers of underwater nodes overlap one another, which may cause an increase in collisions between nodes [58]. In addition, another problem of k-coverage is the high cost of underwater nodes, which is ignored in this study.

Almutairi and Mahfoudh [59] proposed UnderWater Distributed Virtual Force Algorithm (UW-DVFA) to address coverage holes and connectivity issues in 3D network architecture. The main objective of this study was to achieve full coverage and connectivity through random node deployment. To accomplish this, the proposed algorithm requires sensor nodes to redeploy their positions to more sparsely populated or unpopulated areas. Simulation results reveal that the proposed algorithm achieves full coverage when sensor nodes number from 400-500. The results of this study also show that delay decreases as the number of sensor nodes increases. One of the major issues of this study, however, is that the total number of sensor nodes is extremely high relative to the simulation area. Since underwater sensor nodes are extremely expensive, this makes the proposed algorithm inefficient for monitoring real-world underwater applications. Moreover, the authors assumed that by achieving full coverage, connectivity issues would be resolved. This is a false assumption as having sensor nodes cover the whole simulation area does not mean all nodes can communicate with each other. This means there is still the possibility of isolated nodes that cannot communicate with other sensor nodes, which hence degrade the network performance. Shadow zones negatively affect network connectivity and underwater communication as they create signals [60]. Therefore, it is necessary to deploy underwater nodes in such a way that ensures connectivity and minimizes signal transmission loss.

Li et al. [61] first analyzed an AoI for 3D network architecture that involved deploying a sink node to a predetermined location so that it can reach underwater sensors with minimal delay. Underwater sensors are deployed randomly after which each sensor adjusts its depth by adjusting its anchor wire connected to the sensor. The depth adjustment is based on AoI. The authors propose a node deployment algorithm for the sink node where it changes its position according to the column of the underwater sensors. The surface area is broken down into multiple columns. Each column has an underwater sensor that monitors an AoI. To minimize delay, the minimum distance between two nodes is when both are in a straight line. Thus, the proposed node deployment algorithm requires the sink node to change its position to achieve minimum distance and line up with a given underwater sensor. Although the proposed algorithm can minimize delay between a single underwater sensor and the sink node, its requirements increase the delay for other underwater sensors to the sink, due to the sink relocating. To avoid increasing

delay in this way, the underwater sink should be deployed in the middle of an AoI. Or, this model can be utilized but with the use of multiple underwater sink nodes; however, this will increase costs. Another limitation of this study is that it did not address coverage holes, connectivity issues, and other performance metrics.

Zhang et al. [62] proposed a multiple sink node deployment algorithm for 2D network architecture to minimize E2ED and EC. This algorithm relies on Cuckoo optimization algorithm to optimize the location of sink nodes. In addition, this study evaluated the impact of multiple sink nodes in terms of E2ED and EC. The location of the sink nodes plays a crucial role in enhancing underwater communication. However, due to the high cost of sink nodes, it is necessary to require only the minimal number necessary to achieve reduced delay and EC. The results of this study reveal that the proposed algorithm achieves lower E2ED and EC. The simulation results also show that as the number of surface sinks increases, E2ED and EC decrease. However, this study relies on a large number of sensor nodes and sink nodes, which increases the cost. Another problem is that the authors assumed that sensor nodes are directly connected to the nearest sink node through single-hop fashion, which is not the case in most underwater applications. Commonly, underwater sensors transmit collected data to relay nodes to reach the sink node. This means that the proposed algorithm will not be an efficient approach to monitor most underwater applications. Moreover, this study ignores other sensor node deployment algorithms, such as the impact of deploying sensor nodes randomly, coverage holes, connectivity issues, cost, and other metrics of network performance. This study also relies on TDMA MAC protocol, which commonly results in high delay compared to other MAC protocols and hence is not an efficient approach for monitoring time-sensitive applications.

Liu et al. [63] proposed an optimal sensor node deployment for relay nodes and flow allocation called Alternative Flow and Relay-node Adjustment (AFRA) in 3D network architecture for UASNs. UASNs consist of sensor nodes, relay nodes, and sink nodes. Commonly, in 3D and 4D network architectures, data packets get transmitted through relay nodes to reach the sink node. Since underwater nodes are equipped with a very limited battery that is difficult to recharge or replace, this study aims to minimize EC through relocating relay nodes. This means that to extend the network lifetime, relay nodes are utilized. To achieve this, the distance between the sensor nodes, relay nodes, and sink nodes should be reduced. As the distance between nodes decreases, there is lower EC required to transmit packets between intended parties. Therefore, it is critical to find the optimal location of the relay nodes from the sensor nodes, other relay nodes, and the sink nodes, to minimize EC and hence extend the network lifetime. Particularly, this study proposes Relay-node Adjustment algorithm to readjust relay nodes to an optimal location to achieve the above goals. Simulation results reveal that the proposed algorithm can result in higher network lifetime compared to other algorithms. However, since sensor nodes are randomly deployed, coverage holes and connectivity issues may exist and hence this may not be an efficient solution for timesensitive applications. Furthermore, this study relies on a large amount of sensors in a small target area, which increases the cost and the potential of collision.

Kahriman et al. [64] proposed sink node deployment algorithm to maximize coverage for 2D network architecture that only requires a low number of sensors. The placement of sink nodes plays a critical role not only in enhancing coverage, but also network performance. Therefore, it is very critical to deploy sink nodes appropriately to assist in meeting the QoS requirements [65]. The optimal location of the sink node is chosen when a sink node can connect with the highest number of underwater sensors. For example, if the sink node can communicate with 8 sensor nodes in location x and can communicate with 10 sensor nodes in location y, location y is considered the optimal location. The optimal location of the sink nodes is then compared to the positioning of the correlated number of underwater sensors to determine the total number of sensors required to maximize coverage. The study also analyzes the amount of power needed to transmit packets between intended parties. Analyzing the power level involves determining the amount of energy a sensor node must consume to transmit packets to the sink node. This means that the power level can tell us which algorithm can best minimize EC and optimally extend network lifetime. The simulation results reveal that the proposed algorithm achieves higher coverage rate and lower EC compared to a similar algorithm. However, coverage holes and connectivity issues still exist due to the method for locating the sensor nodes. In addition, EC could be further reduced if the authors consider the chance of collisions between sensor nodes.

Nazrul Alam and Haas [18] proposed an SND algorithm to address coverage and connectivity issues in 3D network architecture. This study focused on enhancing coverage and connectivity in a network where sensor nodes are deployed randomly. To achieve full coverage in such a random node deployment, many sensors must be deployed, but only a select number of the nodes need to be active to achieve full coverage. This study aims to choose the appropriate nodes to maintain as active to ensure full coverage while others are put in sleep mode to extend the network lifetime. The main objective of this paper was to analyze the relationship between maximum coverage and maximum connectivity in 3D network architecture. To accomplish this, the proposed algorithm divides the network into virtual cells using truncated octahedron tessellation technique, where only a single sensor node is active in each cell. Specifically, the proposed algorithm requires changing the radius of each cell in order to achieve k-coverage. Simulation results reveal that 2D network architecture outperforms 3D network architecture

Proposal	Year	SND Target	SND Strategy	Network Architecture	Mobility	Coverage	Connectivity	E2ED	EC
[67]	2014	Sen	Random	3D	No	 ✓ 			 ✓
RBVF, VFRBEC [57]	2018	Sen	Random	3D	No	 ✓ 			 ✓
DABVF [20]	2019	Sen	Random	3D	Yes	 ✓ 			 ✓
[59]	2017	Sen	Random	3D	Yes	 ✓ 	√	√	
[61]	2012	Sin	Random	3D	No			√	
[62]	2016	Sin	Deterministic	2D	No			√	 ✓
AFRA [63]	2017	Rel	Deterministic	3D	No			√	 ✓
UW-DFA [18]	2015	Sen	Random	3D	Yes	 ✓ 	√		 ✓
[66]	2019	Sin	Deterministic	2D, 3D	No			√	 ✓
[68]	2009	Sen	Deterministic	3D	No		√		 ✓

TABLE 1. Summary of the design characteristics and common performance metrics of SND algorithms in UASNs.

Sen: Sensor nodes, Rel: Relay nodes, Sin: Sink nodes

in 1-coverage, whereas 3D network architecture results in better coverage than 2D network architecture in k-coverage. In addition, the proposed algorithm uses selected nodes and hence consumes less energy. The main issue with this study is that it requires many sensors in each cell, hence this is not a cost-efficient approach due to the high cost of underwater nodes. One of the fundamental requirements to developing an optimal sensor location algorithm while achieving full coverage and connectivity is that the AoI is monitored with the minimum number of sensors. Since nodes are randomly deployed, another false assumption is that of full connectivity, which may not be achieved. Some nodes may be isolated from others and hence unable to connect to the other sensor nodes in the network. This also indicates that there is a possibility of coverage holes with this proposal. All of the aforementioned issues can impact network performance metrics, such as E2ED, EC, and PDR.

UASNs have been used to monitor different sizes of AoI. Sending a large amount of data from underwater sensors to a single sink may result in high delay. Therefore, Albarakati et al. [66] focused on the optimization of the multiple sink deployment for time-sensitive applications to monitor a large area. Particularly, the optimal location of the sink and relay nodes depends on the type of network architecture to achieve minimum E2ED and EC. The relay node is responsible for forwarding data from sensor nodes to the sink node. In addition, the proposed algorithm utilizes the characteristics of geometric distribution of AoI, which helps to achieve more accurate results for a real-world underwater environment. Simulation results reveal this proposed method can minimize E2ED and EC. However, the proposed method was not compared with other node deployment schemes. Instead, the authors compared their method to different network architectures. In addition, this study deployed underwater nodes randomly, which may result in high delay, coverage holes, and low connectivity. All these issues were ignored in this study. Deploying multiple sink nodes is suitable for a very large area to enhance network performance. However, the simulation area in this study is considered a small area and a single sink node should be sufficient if sensor nodes and relay nodes are deployed at appropriate locations.

There are many factors that play a critical role in resolving SND issues, such as SND target, SND strategy, network architecture, and mobility. In all underwater applications, the main goal is to enhance network performance metrics while maximizing coverage and connectivity and utilizing a low number of sensor nodes. To evaluate SND algorithms properly, it is necessary to first identify which factors and metrics have been considered in the development of an optimal SND algorithm that can meet the QoS requirements of a target application. In Table 1 the design factors and performance metrics that have been used in recent research papers are summarized. A check mark in the table means that the study focused on the given SND issue (e.g., coverage, connectivity, E2ED, EC) or compared the proposed algorithm with other SND algorithms based on specific network performance metrics. As is shown by the table, two network performance metrics are used in the evaluation: E2ED and EC.

As observed from the literature review, most research studies have focused on SND to resolve coverage holes, connectivity issues, and/or enhance selective performance metrics through random node deployment algorithms. Although related research papers enhance these factors, further enhancement can be achieved by considering the location of all types of underwater nodes (e.g., sensor, relay, and sink) through a deterministic distributed node deployment algorithm.

IV. METHODOLOGY

To develop an optimal SND algorithm for a real scenario of a particular underwater application, certain information is collected about the target application [69]. In addition, the researcher focused on ultra-deep-water (depth greater than 1,500 m) where more than one tsunami event has previously occurred [70], [71]. Therefore, in this study, the focus was on an application that monitors for the potential of tsunami in the Solomon Islands. Since 1926, this area has experienced several devastating tsunami events [72].

To start, GIS data, such as depth and simulation area, must be collected through NOAA, GeoMapApp, or Esri [46], [56], [73], as shown in Figure 8. Second, the requirements of the NOAA standard must be met, which require sensor nodes transmit collected data to the sink node every 15 min to assess the potential for a tsunami. In order to meet the requirements of the target application, this study focused on 3D network architecture where multiple nodes are placed at multiple depths. Next, this case study is evaluated based on a real underwater modem. For this, it is necessary to



FIGURE 8. GIS solomon islands [46].

choose an appropriate underwater modem that can offer high data rate while consuming lower energy to extend network lifetime [74]. Using the above information allows for the development of an optimal SND algorithm that can meet the QoS requirements of the target application.

By considering the delay between nodes, packet size, available data rate, network size, network load, transmission range of underwater modem, and coverage of an AoI, the optimal distance between the underwater nodes is chosen to meet the QoS requirements of the tsunami application. The proposed SND scheme aims to minimize E2ED and EC while maximizing coverage and connectivity and minimizing cost using the characteristics of a real underwater modem. Our proposed node deployment algorithm relies on the GIS of the target area. Since this study is focused on detection of a tsunami event, the chosen setting for monitoring for tsunami was the Solomon Islands in the South Pacific Ocean. The delay between nodes depends on the available data rate, where the available data rate depends on the distance between nodes. Long distances between nodes result in lower available bandwidth, whereas short distances between nodes can offer higher data rates. Higher data rate results in lower delay and vice versa. To minimize delay and EC, the results suggested utilizing a higher data rate, small network size, small packet size, and low network load. In addition, since the cost of underwater nodes is extremely expensive compared to the cost of nodes in TWSNs [10], [29], it was necessary in this study to minimize the total number of underwater nodes utilized. The required number of sensor nodes increases as the AoI increases. In addition, the minimum number of nodes required to fully monitor the AoI relies on the sensing range.

Although higher overlapping areas can cause interference, these can also increase network reliability when a sensor node fails. High network reliability and low interference can be achieved by scheduling data transmissions properly. In such cases, overlapping will not be an issue. Most studies view overlap as always having a negative impact on network performance, but this is actually dependent on the amount of overlap between neighboring nodes and the technique of data transmission employed. Overlap between neighboring nodes can be handy to ensure reliability in case of sensor node failure. In other words, overlap between neighboring nodes
 TABLE 2. Required information for designing suitable SND of target application.

Type of Information	Description			
Target Application	Determine the target underwater application is considered as a real-time or non-real-time application.			
Network Architecture	Select the appropriate network architecture (1D, 2D, 3D, or 4D) that is suitable to address the targeted issues.			
Target Area	Collect GIS information about the AoI, such as depth level (shallow-water, deep-water, ultra-deep-water), topographic information, and size.			
Characteristics of Underwater Nodes	Determine the available data rate, sensing range, and communication range of underwater nodes.			
Communication Reliability	Determine how the flow of data from under- water nodes can reach the sink node (single- hop single path, single-hop multi-path, multi- hop single-path, or multi-hop multi-path).			
Data Reporting System	Determine how data must be reported (event- driven or on-demand).			
Mobility	Determine whether underwater nodes must be static or mobile to meet QoS requirements.			
Coverage and	Determine how underwater nodes are			
Connectivity	interconnected (1-coverage, k-coverage,			
Techniques	1-connectivity, k-connectivity).			
Network setup	Determine the network size, packet size, and network load that is suitable to meet the re- quirements of the target application.			

can be useful to broadcast packets through multi-hop multipath networks. Therefore, it is necessary to have overlap between nodes to ensure reliability and maximize network performance in case of node failure. Table 2 lists the required information to propose a suitable SND algorithm that meets the QoS requirements of the target application.

In this work, underwater nodes are deployed with the focus being to meet the QoS requirements of the time-sensitive application while resolving coverage and connectivity issues. Particularly, underwater nodes are deployed in such a way so as to reduce the interference level and hence enhance network performance. In this work, a DDOGDA for 3D network architecture is developed in such a way so as to meet the QoS requirements of the target application. The main aim of this algorithm is to deploy underwater nodes in locations where the interference level is minimized. Minimizing the interference level can minimize delay and EC, which are critical metrics to meet the QoS requirements of such timesensitive applications.

To deploy underwater nodes properly, the proposed algorithm divides the AoI into grids. First, an underwater sink is deployed in the middle of the AoI so that E2ED can be minimized. Next, each underwater modem is deployed one by one to ensure full coverage and connectivity. Each node must be deployed within the target AoI to ensure that deployed underwater nodes can monitor the whole AoI. If any node has been deployed outside the target AoI, it must relocate itself. Once a node has been deployed within the target AoI, the number of neighbors within the Sensing Range (SR) of each node must be discovered. Each node must have

a neighbor to avoid node isolation. Next, each node must cover a unique area within the grid target area to avoid two nodes covering the same area and wasting network resources. Once the nodes meet the above criteria, this ensures the position of nodes with k-connectivity, avoids node isolation, and confirms 1-coverage. To double check that the chosen position of a node is appropriate, each node must transmit data packets and the sink node must receive these transmitted packets with no loss. If a node within the SR can reach only one node among a number of nodes, it must relocate itself to avoid node isolation. Similarly, if a node fails to cover a unique area within the target area, it must relocate itself to ensure k-connectivity and 1-coverage. When a node transmits data packets toward the sink node that cannot be received by the sink node, this indicates that the position of the node is not reachable and hence it must relocate itself. This process repeats until all underwater nodes have been deployed in an appropriate location within the AoI. The DDOGDA algorithm process is shown in Algorithm 1. Table 3 defines the meaning of each symbol in Algorithm 1.

The AoI of interest is divided into nine cells. In each cell, one only sensor must exist to achieve 1-coverage and hence minimize collisions, which helps to extend network lifetime by minimizing the EC. In our proposed algorithm, the sensing range is assumed to be half of the communication range to ensure full connectivity. This way, each sensor can connect with the next hop node on the same depth level and the node on the lower depth level as they are both within its SR. To further increase communication reliability, each sensor can reach other nodes using its Communication Range (CR). Figure 9 shows how underwater nodes are deployed and communicate with each other using the proposed node deployment algorithm.

The location of sensor nodes and relay nodes is determined according to the simulation area from NOAA to monitor for a potential tsunami for the Solomon Islands. Both sensor nodes and relay nodes can reach the sink node through single-hop or multi-hop, depending on the location of the underwater nodes. In addition, sensor nodes and relay nodes can transmit packets to the sink node via single-path or multipath. Underwater nodes are deployed at different depths to monitor the entire target area. Since the location of the underwater nodes is predetermined, this study assumes that underwater nodes do not move. This can be achieved by anchoring the nodes to ensure the underwater network is a static network. This study assumed that the sink node is not energy constrained due to its ability to use renewable energy sources (e.g., solar energy), while the sensor and relay nodes have limited energy and cannot be recharged. Since our proposed system depends on 3D network architecture, a lower power level is needed and hence this helps to extend the network lifetime and minimize E2ED.

V. EVALUATION AND RESULTS

This study analyzes the performance of underwater communication based on the transmitted data from source

Algorithm 1 DDOGDA Algorithm Process

Objective: To deploy underwater nodes properly, in a manner that improves key factors of performance metrics to meet the QoS requirements of the target application. Input: Node_i, D_p Output: R_p, Pos_i **Initialize:** $Pos_i = 0$, $T_{sn} = 9$, $R_n = 0$ while $Node_i \in T_{sn}$ do Set the initial position of Node_i if $Pos_i \in G_{ta}$ then Count N_n **if** N_n within $S_r > 1$ & Pos_i cover a unique area within G_{ta} then Connectivity = k-connectivity Avoid node isolation Coverage = 1-coverage Node_{*i*} transmits D_p to S_n **if** D_p of Node_i can reach S_n **then** Set Node_{*i*} = Pos_i else Change the initial position of Node_i if Ack_p is received then R_p ++ else Retransmit L_n end end else Change the initial position of Node_i end else Change the initial position of Node_{*i*} end end Report Pos_i Report R_n

TABLE 3. DDOGDA reference table.

Symbol	Description
Node _i	represents the source node i where <i>i</i> refers to
	$\{1, 2,n\}.$
D_p	represents the data packets.
R_p	represents that the sink node has received the
	transmitted data packets.
Pos _i	represents the position of node i.
T _{sn}	represents the total number of source nodes.
G _{ta}	represents the grid target area.
N _n	represents the number of neighbor nodes.
S_r	represents sensing range.
Sn	represents the sink node.
Ack _p	represents the acknowledgment packet.
L_p	represents lost packets.

nodes toward the sink node. To save lives, the tsunami monitoring application requires minimum delay and maximized PDR. To achieve the QoS requirements of this type of application, different node deployment algorithms must



FIGURE 9. Proposed node deployment algorithm.

be investigated. To optimize the underwater communication in the tsunami application, the impact of different node deployment algorithms must be analyzed through common network performance metrics. Based on the analysis of the literature review, many recent papers have addressed the performance of underwater communication issues through random [6], [22], [23], [36], [44], [58], [63], [66], tetrahedron [11], [18], [44], cuboid [11], [18], [29], [32], [44], triangular [18], [35], [36], pipeline [6], and grid node deployment algorithms. Therefore, this research evaluates the proposed algorithm compared to other node deployment algorithms in terms of E2ED, EC, and PDR to prove how the proposed algorithm can further enhance the performance of underwater communication.

A. SIMULATION SETUP

This study utilized the Aqua-Sim tool, which is an NS-2 simulator for evaluating different node deployment algorithms [75], [76]. To assess the network performance of the proposed, random, tetrahedron, cuboid, triangular, pipeline, and grid node deployment algorithms in terms of E2ED, EC, and PDR, the network is set up as an ad-hoc network topology so that meeting the QoS requirements of the target application is possible. In this network topology, some underwater nodes must rely on the multi-hop fashion to transmit data packets to the sink node due to the large distance between sensor nodes and the sink node. This study relies on homogeneous nodes where all nodes have the same characteristics in terms of data rate, energy consumption, operating frequency, communication range, SR, etc. In this study, all underwater nodes are stationary and deployed in a specific location according to the proposed algorithm. In the random node deployment algorithm, underwater nodes are deployed randomly. This study relies on a random function to choose random locations within the AoI for the random node deployment algorithm. In the tetrahedron, cuboid, triangular, and pipeline node deployment algorithms, underwater nodes are deployed deterministically in the AoI to shape tetrahedron, cuboid, triangular, and pipeline. In the grid node deployment algorithm, underwater nodes are deployed

Parameters	Value		
Radio propagation model	Underwater Propagation		
Channel	Underwater Channel		
Routing protocol	Vector-based Forward (VBF)		
Number of sensor nodes	9		
Number of sink nodes	1		
Simulation area	$2412 \text{ m} \times 1807 \text{ m}$		
Simulation time	3600 s		
Initial energy	10,000 J		
Transmission power	2.0 W		
Receiving power	0.8 W		
Idle power	0.008 W		
Operating frequency	26.77 kHz - 44.62 kHz		
Data rate	17.8 Kbps		
Network load	0.05 packets/sec		
Packet size	60 Bytes		
Control packet size	20 Bytes		
Type of traffic	Constant Bit Rate (CBR)		
Communication range	1200 m		
Sensing range	600 m		
Operating frequency Data rate Network load Packet size Control packet size Type of traffic Communication range Sensing range	26.77 kHz - 44.62 kHz 17.8 Kbps 0.05 packets/sec 60 Bytes 20 Bytes Constant Bit Rate (CBR) 1200 m 600 m		

in random locations on each grid. Table 4 illustrates all simulation parameters utilized in this study [74], [77].

B. PERFORMANCE METRICS

This study evaluated the proposed algorithm compared to the random, tetrahedron, cuboid, triangular, pipeline, and grid node deployment algorithms in terms of E2ED, EC, and PDR. The details of these network performance metrics are described in the following:

 End-to-End Delay (E2ED): is the total time each packet takes from source nodes to destination node divided by total number of received data packets. This is measured in seconds (s). E2ED can be calculated using:

$$E2ED[s] = \frac{TD}{TR} \tag{1}$$

where, TD = total delay of all packets sent from source nodes to sink node; TR = total number of received packets from source nodes to sink node.

2) Energy Consumption (EC) is the sum of the energy consumed by all nodes, including transmitted power (E_{tx}) , receiving power (E_{rx}) , and idle power (E_{idl}) , to transmit packets between parties. This is measured in joules (J). EC can be calculated using:

$$EC[J] = \sum_{i=1}^{n} (E_{tx} + E_{rx} + E_{idl})$$
(2)

where, n = total number of nodes in the network.

3) Packet Delivery Ratio (PDR) can be defined as the ratio of the total number of received packets at the sink node (TR_{sn}) to the total number of sent packets (TS). PDR can be calculated using:

$$PDR = \left(\frac{TR_{sn}}{TS}\right) \times 100 \tag{3}$$



FIGURE 10. End-to-end delay.



FIGURE 11. Energy consumption.



FIGURE 12. Packet delivery ratio.

C. RESULTS AND ANALYSIS

In this section, first the network performance of the DDOGDA algorithm is evaluated and then it is compared to six other different node deployment algorithms: random, tetrahedron, cuboid, triangular, pipeline, and grid. In particular, this study compares these algorithms in terms of E2ED (see Figure 10), EC (see Figure 11), and PDR (see Figure 12). Since this study relies on 3D network architecture, it utilizes a multi-hop network with nine sensor and relay nodes laid underwater and a single sink node at the surface level. The underwater nodes rely on a multi-hop network so packets from sensor nodes can reach the sink node through

relay nodes. To ensure a fair comparison, it is critical to consider the coverage and connectivity ratio of the DDOGDA and other SND algorithms. Simulation results show that the DDOGDA algorithm results in maximum coverage, while other algorithms could not resolve the coverage holes. Similarly, the connectivity ratio of the DDOGDA, cuboid, and grid algorithms offer the maximum connectivity ratio while the connectivity issues remain a major problem in the random, tetrahedron, triangular, and pipeline algorithms. In terms of both coverage and connectivity, the DDOGDA offers the highest results in comparison to other algorithms. This helps to ensure that the DDOGDA can monitor the entire AoI and hence can generate an early warning for evacuation in the event of a tsunami more effectively.

Figure 10 shows the E2ED of different SND algorithms using the DDOGDA and compares the results to those of the random, tetrahedron, cuboid, triangular, pipeline, and grid node deployment algorithms. The position of underwater nodes in any SND algorithm has a great impact on E2ED. The larger the distance between underwater nodes, the higher the E2ED, which as a result may endanger human lives. In addition, to place underwater nodes properly, the location of the sensor, relay, and sink nodes must be all taken into account. Deploying only sensor nodes properly and ignoring relay and sink nodes does not guarantee better network performance due to potential communication issues, such as coverage holes and connectivity issues. Therefore, the lower the E2ED, the higher the chance to save lives by generating an early warning for evacuation. This means that in order to minimize E2ED, it is recommended that researchers deploy underwater nodes close to one another while addressing other SND issues, such as coverage holes and connectivity issues. The E2ED varies depending on the location of the underwater nodes, where the DDOGDA algorithm offers the lowest delay compared to other node deployment algorithms. The E2ED delay in DDOGDA, random, tetrahedron, cuboid, triangular, pipeline, and grid is as follows: 1.03289566 s, 2.136529944 s, 2.744311725 s, 1.353626679 s, 3.780817695s, 2.614103777s, 2.474413531s, respectively. This shows that the DDOGDA can transfer packets to the sink node with the minimum delay in comparison to other algorithms and therefore can save human lives by generating an early warning for evacuation. In particular, DDOGDA reduces the E2ED as compared to random, tetrahedron, cuboid, triangular, pipeline, and grid node deployment algorithms by about 106.85%, 165.69%, 31.05%, 266.04%, 153.05%, and 139.56%, respectively. This shows that the proposed algorithm can deliver packets much faster than other SND algorithms, which again correlates with greater effectiveness in generating an early warning for evacuation and saving lives. This means that delivering the packets with minimum delay can assist in avoiding the potential of catastrophic events resulting in fatalities.

Since the underwater nodes can be deployed either randomly or deterministically, neither can guarantee lower E2ED in any target application. This means that choosing

the appropriate node deployment algorithm is necessary to meet the requirements of the target application. The DDOGDA, cuboid, tetrahedron, triangular, and pipeline node deployment algorithms rely on deterministic node deployment and their results vary. For example, the results of DDOGDA and cuboid node deployment algorithms provide the lowest E2ED while pipeline, tetrahedron, and random and grid deploy underwater nodes randomly. Based on the results of E2ED, the following node deployment algorithms are ranked from lowest to highest E2ED: DDOGDA, cuboid, random, grid, pipeline, tetrahedron, and triangular, respectively. This indicates that the DDOGDA outperforms the other node deployment algorithms in terms of E2ED and hence is more suitable for time-sensitive applications, such as flood, volcanic eruption, tsunami, earthquake, pipeline malfunction/break/leak, and military. The results above also show that random and grid node deployment algorithms outperform other deterministic node deployment algorithms, such as the pipeline, tetrahedron, and triangular. Moreover, the findings of this research can assist future researchers in selecting the appropriate node deployment algorithms for meeting the requirements of their target applications.

Figure 11 shows the total EC of DDOGDA and compares the results to those of the random, tetrahedron, cuboid, triangular, pipeline, and grid node deployment algorithms. The EC differs from one SND algorithm to the next as each one has its unique way of transmitting packets to intended parties. A higher number of packets received results in higher EC and vice versa. Similarly, higher collision rate results in higher EC and vice versa. As can be observed in Figure 11, the DDOGDA consumes the smallest amount of energy followed by the tetrahedron, cuboid, grid, triangular, random, and pipeline node deployment algorithms. In particular, the DDOGDA consumes 820.593129 joules while the tetrahedron, cuboid, grid, triangular, random, and pipeline consume 846.33507 J, 1308.156346 J, 1447.741514 J, 1720.318987 J, 2248.079066 J, 2323.312618 J, respectively. This shows that the DDOGDA consumes the least energy among the examined SND algorithms and hence can extend network lifetime in comparison to the tetrahedron, cuboid, grid, triangular, random, and pipeline node deployment algorithms by about 3.14%, 59.42%, 76.43%, 109.64%, 173.96%, and 183.13%, respectively. This shows that the DDOGDA algorithm achieves the lowest EC in comparison to the other SND algorithms and hence can maximize network lifetime, which makes it more appropriate to monitor time-sensitive applications. Moreover, extending the network lifetime of the UASN makes it more reliable and efficient by reducing its total cost. Based on the above results, it is clear that when underwater nodes are deployed deterministically, such as with DDOGDA, tetrahedron, and cuboid node deployment algorithms, less energy is consumed than when the nodes are deployed randomly, such as with grid and random node deployment algorithms. Similarly to the E2ED, not all node deployment algorithms that utilize deterministic techniques for underwater nodes can guarantee lower EC than those

that use random techniques. The above results show that the grid node deployment algorithm consumes less energy than the triangular node deployment algorithm. Similarly, the random node deployment algorithm consumes lower EC than the pipeline node deployment algorithm. Therefore, it is necessary to collect information about the target application, including the GIS data of its target area and other information, to determine the most appropriate SND algorithm.

To decide the best SND algorithm in this study, it is critical to determine the type of target application (i.e., time-sensitive or non-time-sensitive). Since this research focuses on an application used for monitoring tsunamis, which is clearly a time-sensitive application, minimizing EC is not a high priority in comparison to optimizing E2ED and PDR. In the case of a disaster-monitoring application, the SND algorithm is less valuable if reported data from underwater nodes takes a very long time to reach the sink node or if some of the reported data does not reach the sink node. This endangers humans' lives and hence does not meet the requirements of the target application. Therefore, for time-sensitive applications, E2ED and PDR are given higher priority over EC. However, enhancing E2ED and PDR while also consuming lower EC is indispensable, as underwater nodes are costly and have a limited battery. Once the battery of an underwater node is drained, it must be replaced as it cannot be recharged. In this study, the DDOGDA consumes the lowest E2ED and EC in comparison to cuboid and grid node deployment algorithms, which all offer maximum PDR. Since the proposed algorithm can outperform other SND algorithms on multiple factors, this study concludes that the proposed algorithm is the most suitable for time-sensitive monitoring applications.

In Figure 12, the PDR of the DDOGDA algorithm is presented and compared to the results of the random, tetrahedron, cuboid, triangular, pipeline, and grid node deployment algorithms. The PDR result varies depending on the node deployment strategy utilized. A higher PDR is one of the fundamental requirements to ensure that all generated packets are successfully delivered to the sink node. It can be observed from Figure 12 that the DDOGDA, cuboid, and grid node deployment algorithms achieve higher PDR compared to the random, tetrahedron, triangular, and pipeline node deployment algorithms. Due to their sparse node deployment, random, tetrahedron, triangular, and pipeline algorithms are not able to deliver large numbers of packets to the sink node. This indicates that these other SND algorithms experience coverage holes, connectivity issues, and/or a large number of Total Collisions (TC), which indicates they are not suitable for time-sensitive monitoring applications. To maximize the PDR in the random, tetrahedron, triangular, and pipeline algorithms requires additional nodes, which increases the total cost of the system. Although the DDOGDA, cuboid, and grid algorithms result in higher PDR compared to the other SND algorithms, the DDOGDA protocol achieves the maximum PDR while minimizing E2ED and EC, in comparison to the cuboid and grid algorithms. This means that the DDOGDA algorithm again outperforms the other SND algorithms.

In particular, the DDOGDA, cuboid, and grid algorithms achieve 100% while the random, tetrahedron, triangular, and pipeline achieve 88.85%, 88.89%, 77.78%, 87.47%, respectively. One of the main reasons that these node deployment algorithms did not offer maximum PDR is that in their scenarios some packets from underwater nodes cannot reach the sink node due to node isolation and/or packet collisions.

Moreover, the DDOGDA offers maximal PDR in comparison to that of the random, tetrahedron, triangular, and pipeline algorithms due to their coverage issues, by about 11.15%, 11.11%, 22.22%, and 12.53%, respectively. This indicates that the DDOGDA is considerably more suitable for tsunami monitoring underwater applications. The proposed algorithm can deliver all packets transmitted from underwater nodes while other SND algorithms miss some packets. In the context of disaster warning, missing some packets might send authorities incorrect data on what appropriate actions must be taken in a given situation, which could endanger citizens and delay alerts regarding catastrophic hazards. Therefore, it is a priority that all packets from underwater nodes to the sink node are delivered so that appropriate action can be taken. The above results show that the highest PDR is offered by two algorithms, the proposed and the cuboid, which rely on deploying underwater nodes deterministically. However, deploying underwater nodes using deterministic technique does not guarantee better PDR than those of the one relying on the random technique to meet the requirements of any target applications. The above result shows that some random algorithms outperform deterministic ones. In particular, the grid algorithm achieves higher PDR than tetrahedron by about 11.11%. Similarly, the random algorithm offers higher PDR than the pipeline and triangular algorithms by about 1.56% and 12.46%, respectively. Nevertheless, the deterministic approach has a higher chance of meeting the requirements of the target applications by requiring a lower number of underwater nodes and hence costing less.

VI. CONCLUSION

In this paper, the author develops a DDOGDA algorithm for a time-sensitive application in UASNs. The proposed DDOGDA relies on a deterministic node deployment mechanism to address coverage holes and connectivity issues while meeting the QoS requirements. This study confirms that the DDOGDA should be considered the most appropriate algorithm for use with a time-sensitive monitoring application designed to save human lives by generating an early warning for evacuation, due to its E2ED, EC, and PDR. The DDOGDA addresses the coverage holes, connectivity issues, and low network performance by utilizing an efficient node deployment scheme that improves underwater communication. Moreover, this study compared the results of the DDOGDA algorithm to those of the random, tetrahedron, cuboid, triangular, pipeline, and grid node deployment algorithms to highlight the differences between them. In particular, the DDOGDA is the most appropriate for a tsunami-monitoring underwater applications, followed by the cuboid, random, grid, pipeline, tetrahedron, and triangular algorithms, respectively. Although the cuboid and grid algorithms achieve maximum PDR, they require higher E2ED than DDOGDA, do not cover the entire AoI, and consume much higher EC due to a high number of collisions. In short, the proposed DDOGDA algorithm presents significant enhancements in comparison to other SND algorithms in terms of E2ED, EC, and PDR, which makes it the most suitable for tsunami-monitoring underwater applications. This study can serve as a guide for other researchers on how to meet the QoS requirements of target applications.

VII. FUTURE WORK

Since this study focused on stationary nodes, further investigation is needed to evaluate the following scenarios: mobile network, non-delay sensitive applications, 1D network architecture, 2D network architecture, 4D network architecture, single-hop single path, single-hop multi-path, multi-hop single-path, k-coverage, 1-connectivity, and largescale network. Further investigation is also needed to evaluate different characteristics of underwater modems. In particular, further improvement of this research can be achieved by investigating higher data rates, communication range, and sensing range to deliver packets further with a lower number of collisions and hence minimize delay and energy consumption. Moreover, the DDOGDA algorithm needs to be evaluated in different underwater application domains and other network performance metrics. This research can also be extended to other underwater applications, such as military surveillance, environmental monitoring, pollution monitoring, oceanographic data compilation, assisted navigation, mine detection, and oil/gas spill detection. This study can also assist in other domains, such as the Internet of Things (IoT), WSN, and underground communication in monitoring environmental conditions and reporting potential disaster more reliably and efficiently. In particular, this research can assist authorities and industries in appropriate decision-making based on the current status of weather conditions, air quality, climate monitoring, and natural hazards. This functionality can also be extended to collect data from swarm drones that monitor critical areas to the ground station.

REFERENCES

- [1] K. M. Awan, P. A. Shah, K. Iqbal, S. Gillani, W. Ahmad, and Y. Nam, "Underwater wireless sensor networks: A review of recent issues and challenges," *Wireless Commun. Mobile Comput.*, vol. 2019, pp. 1–20, Jan. 2019. [Online]. Available: https://www.hindawi.com/ journals/wcmc/2019/6470359/
- [2] M. Liu, W. Huang, L. Qian, and S. Zhang, "An improved R-MAC based MAC protocol for underwater acoustic networks," in *Proc. IEEE Int. Conf. Signal Process., Commun. Comput. (ICSPCC)*, Aug. 2016, pp. 1–6.
- [3] S. H. Park, P. D. Mitchell, and D. Grace, "Reinforcement learning based MAC protocol (UW-ALOHA-QM) for mobile underwater acoustic sensor networks," *IEEE Access*, vol. 9, pp. 5906–5919, 2021.

- [4] J.-Y. Lee, N.-Y. Yun, S. Muminov, S.-Y. Shin, Y.-S. Ryuh, and S.-H. Park, "A focus on practical assessment of MAC protocols for underwater acoustic communication with regard to network architecture," *IETE Tech. Rev.*, vol. 30, no. 5, pp. 375–381, Sep. 2013. [Online]. Available: https://www.tandfonline.com/doi/abs/10.4103/0256-4602. 123119
- [5] N. Goyal, M. Dave, and A. K. Verma, "Protocol stack of underwater wireless sensor network: Classical approaches and new trends," *Wireless Pers. Commun.*, vol. 104, no. 3, pp. 995–1022, Feb. 2019, doi: 10.1007/s11277-018-6064-z.
- [6] R. A. Khalil and N. Saeed, "Optimal relay placement in magnetic induction-based Internet of Underwater Things," *IEEE Sensors J.*, vol. 21, no. 1, pp. 821–828, Jan. 2021.
- [7] P. Si, J. Ma, F. Tao, Z. Fu, and L. Shu, "Energy-efficient barrier coverage with probabilistic sensors in wireless sensor networks," *IEEE Sensors J.*, vol. 20, no. 10, pp. 5624–5633, May 2020.
- [8] Y. Song and P.-Y. Kong, "Optimizing design and performance of underwater acoustic sensor networks with 3D topology," *IEEE Trans. Mobile Comput.*, vol. 19, no. 7, pp. 1689–1701, Jul. 2020.
- [9] N. Nowsheen, G. Karmakar, and J. Kamruzzaman, "An adaptive approach to opportunistic data forwarding in underwater acoustic sensor networks," in *Proc. IEEE 13th Int. Symp. Netw. Comput. Appl.*, Aug. 2014, pp. 229–236.
- [10] A. Roy and N. Sarma, "Factors affecting MAC protocol performance in underwater wireless sensor networks," *Int. J. Comput. Appl.*, vol. 169, no. 5, pp. 36–41, Jul. 2017. [Online]. Available: http://www.ijcaonline.org/ archives/volume169/number5/27985-2017914755
- [11] G. Han, C. Zhang, L. Shu, and J. J. P. C. Rodrigues, "Impacts of deployment strategies on localization performance in underwater acoustic sensor networks," *IEEE Trans. Ind. Electron.*, vol. 62, no. 3, pp. 1725–1733, Mar. 2015.
- [12] J. Lloret, "Underwater sensor nodes and networks," Sensors, vol. 13, no. 9, pp. 11782–11796, Sep. 2013. [Online]. Available: https://www.mdpi.com/1424-8220/13/9/11782
- [13] M. Hamidzadeh, N. Forghani, and A. Movaghar, "A new hierarchal and scalable architecture for performance enhancement of large scale underwater sensor networks," in *Proc. IEEE Symp. Comput. Informat.*, Mar. 2011, pp. 520–525.
- [14] D. Arivudainambi, S. Balaji, and T. S. Poorani, "Sensor deployment for target coverage in underwater wireless sensor network," in *Proc. Int. Conf. Perform. Eval. Modeling Wired Wireless Netw. (PEMWN)*, Nov. 2017, pp. 1–6.
- [15] F. Han, X. Liu, I. I. Mohamed, K. H. Ghazali, and Y. Zhao, "A survey on deployment and coverage strategies in three-dimensional wireless sensor networks," in *Proc. 8th Int. Conf. Softw. Comput. Appl.* New York, NY, USA: Association for Computing Machinery, Feb. 2019, pp. 544–549, doi: 10.1145/3316615.3316716.
- [16] Z. Jin, Z. Ji, Y. Su, S. Li, and B. Wei, "A deployment optimization mechanism using depth adjustable nodes in underwater acoustic sensor networks," in *Proc. OCEANS, MTS/IEEE Kobe Techno-Oceans (OTO)*, May 2018, pp. 1–6.
- [17] D. Pompili, T. Melodia, and I. F. Akyildiz, "Deployment analysis in underwater acoustic wireless sensor networks," in *Proc. 1st ACM Int. Workshop Underwater Netw. (WUWNet).* New York, NY, USA: Association for Computing Machinery, 2006, pp. 48–55, doi: 10.1145/1161039. 1161050.
- [18] S. M. N. Alam and Z. J. Haas, "Coverage and connectivity in three-dimensional networks with random node deployment," Ad Hoc Netw., vol. 34, pp. 157–169, Nov. 2015. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S1570870514001978
- [19] P. Kumar, P. Kumar, P. Priyadarshini, and Srija, "Underwater acoustic sensor network for early warning generation," in *Proc. Oceans*, Oct. 2012, pp. 1–6. [Online]. Available: https://www.linkedin.com/in/srijas/
- [20] C. Liu, Z. Zhao, W. Qu, T. Qiu, and A. K. Sangaiah, "A distributed node deployment algorithm for underwater wireless sensor networks based on virtual forces," *J. Syst. Archit.*, vol. 97, pp. 9–19, Aug. 2019. [Online]. Available: https://www.sciencedirect. com/science/article/pii/S1383762118305794
- [21] P. Jiang, X. Wang, and L. Jiang, "Node deployment algorithm based on connected tree for underwater sensor networks," *Sensors*, vol. 15, no. 7, pp. 16763–16785, Jul. 2015. [Online]. Available: https://www.mdpi.com/ 1424-8220/15/7/16763

- [22] K. Sathish, R. C. Venkata, R. Anbazhagan, and G. Pau, "Review of localization and clustering in USV and AUV for underwater wireless sensor networks," *Telecom*, vol. 4, no. 1, pp. 43–64, Jan. 2023. [Online]. Available: https://www.mdpi.com/2673-4001/4/1/4
- [23] S. S. Mohar, S. Goyal, and R. Kaur, "Optimized sensor nodes deployment in wireless sensor network using bat algorithm," *Wireless Pers. Commun.*, vol. 116, no. 4, pp. 2835–2853, Feb. 2021, doi: 10.1007/s11277-020-07823-z.
- [24] M. Jouhari, K. Ibrahimi, H. Tembine, and J. Ben-Othman, "Underwater wireless sensor networks: A survey on enabling technologies, localization protocols, and Internet of Underwater Things," *IEEE Access*, vol. 7, pp. 96879–96899, 2019.
- [25] M. R. Bharamagoudra and S. K. S. Manvi, "Deployment scheme for enhancing coverage and connectivity in underwater acoustic sensor networks," *Wireless Pers. Commun.*, vol. 89, no. 4, pp. 1265–1293, Aug. 2016, doi: 10.1007/s11277-016-3315-8.
- [26] H. Keshmiri and H. Bakhshi, "A new 2-phase optimization-based guaranteed connected target coverage for wireless sensor networks," *IEEE Sensors J.*, vol. 20, no. 13, pp. 7472–7486, Jul. 2020.
- [27] S. Chakraborty, N. K. Goyal, and S. Soh, "On area coverage reliability of mobile wireless sensor networks with multistate nodes," *IEEE Sensors J.*, vol. 20, no. 9, pp. 4992–5003, May 2020.
- [28] J. Amutha, S. Sharma, and J. Nagar, "WSN strategies based on sensors, deployment, sensing models, coverage and energy efficiency: Review, approaches and open issues," *Wireless Pers. Commun.*, vol. 111, no. 2, pp. 1089–1115, Mar. 2020, doi: 10.1007/s11277-019-06903-z.
- [29] J. Luo, Y. Yang, Z. Wang, and Y. Chen, "Localization algorithm for underwater sensor network: A review," *IEEE Internet Things J.*, vol. 8, no. 17, pp. 13126–13144, Sep. 2021.
- [30] Z. Yingying, L. Xia, and F. Shiliang, "Deployment analysis in twodimensional underwater acoustic wireless sensor networks," in *Proc. IEEE Int. Conf. Signal Process., Commun. Comput. (ICSPCC)*, Sep. 2011, pp. 1–5.
- [31] D. Zeng, X. Wu, Y. Wang, H. Chen, K. Liang, and L. Shu, "A survey on sensor deployment in underwater sensor networks," in *Advances in Wireless Sensor Networks* (Communications in Computer and Information Science), L. Sun, H. Ma, and F. Hong, Eds. Berlin, Germany: Springer, 2014, pp. 133–143.
- [32] W. Shen, C. Zhang, and J. Shi, "Weak k-barrier coverage problem in underwater wireless sensor networks," *Mobile Netw. Appl.*, vol. 24, no. 5, pp. 1526–1541, Oct. 2019, doi: 10.1007/s11036-019-01273-z.
- [33] M. Torshizi and M. J. Sheikhzadeh, "Optimum K-coverage in wireless sensor network with no redundant node by cellular learning automata," *Wireless Pers. Commun.*, vol. 110, no. 2, pp. 545–562, Jan. 2020, doi: 10.1007/s11277-019-06741-z.
- [34] Y. Yue, L. Cao, and Z. Luo, "Hybrid artificial bee colony algorithm for improving the coverage and connectivity of wireless sensor networks," *Wireless Pers. Commun.*, vol. 108, no. 3, pp. 1719–1732, Oct. 2019, doi: 10.1007/s11277-019-06492-x.
- [35] W. Kim, H. W. Moon, and Y. J. Yoon, "Adaptive triangular deployment of underwater wireless acoustic sensor network considering the underwater environment," *J. Sensors*, vol. 2019, pp. 1–11, Feb. 2019. [Online]. Available: https://www.hindawi.com/journals/js/2019/6941907/
- [36] M. Alsulami, R. Elfouly, and R. Ammar, "Underwater wireless sensor networks: A review," in *Proc. 11th Int. Conf. Sensor Netw.* Setúbal, Portugal: SciTePress, 2022, pp. 202–214.
- [37] S. Fattah, A. Gani, I. Ahmedy, M. Y. I. Idris, and I. A. T. Hashem, "A survey on underwater wireless sensor networks: Requirements, taxonomy, recent advances, and open research challenges," *Sensors*, vol. 20, no. 18, p. 5393, Sep. 2020. [Online]. Available: https://www.mdpi.com/1424-8220/20/18/5393
- [38] S. Ibrahim, J. Liu, M. Al-Bzoor, J.-H. Cui, and R. Ammar, "Towards efficient dynamic surface gateway deployment for underwater network," *Ad Hoc Netw.*, vol. 11, no. 8, pp. 2301–2312, Nov. 2013. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S1570870513001200
- [39] J. J. Kartha and L. Jacob, "Network lifetime-aware data collection in underwater sensor networks for delay-tolerant applications," *Sādhanā*, vol. 42, no. 10, pp. 1645–1664, Oct. 2017, doi: 10.1007/s12046-017-0713-x.
- [40] S. Debnath and A. Hossain, "Network coverage in interference limited wireless sensor networks," *Wireless Pers. Commun.*, vol. 109, no. 1, pp. 139–153, Nov. 2019, doi: 10.1007/s11277-019-06555-z.

- [41] M. Choudhary and N. Goyal, "Node deployment strategies in underwater wireless sensor network," in *Proc. Int. Conf. Adv. Comput. Innov. Technol. Eng. (ICACITE)*, Mar. 2021, pp. 773–779.
- [42] F. Aznoli and N. J. Navimipour, "Deployment strategies in the wireless sensor networks: Systematic literature review, classification, and current trends," *Wireless Pers. Commun.*, vol. 95, no. 2, pp. 819–846, Jul. 2017, doi: 10.1007/s11277-016-3800-0.
- [43] S. Abdollahzadeh and N. J. Navimipour, "Deployment strategies in the wireless sensor network: A comprehensive review," *Comput. Commun.*, vols. 91–92, pp. 1–16, Oct. 2016. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0140366416302407
- [44] J. Song, H. Jin, X. Shen, and S. Zhang, "A maximum localization rate algorithm for 3D large-scale UWSNs," *IEEE Access*, vol. 10, pp. 111962–111973, 2022.
- [45] Y. Wang, C. Jiménez, M. Quiroz, and E. Ortega, "Tsunami early warning system using offshore tsunameters in Peru," *Ocean Eng.*, vol. 279, Jul. 2023, Art. no. 114516. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0029801823009009
- [46] National Oceanic and Atmospheric Administration. Accessed: Jul. 3, 2023. [Online]. Available: http://www.noaa.gov/
- [47] T. Melodia, H. Kulhandjian, L.-C. Kuo, and E. Demirors, "Advances in underwater acoustic networking," in *Mobile Ad Hoc Networking*. Hoboken, NJ, USA: Wiley, 2013, pp. 804–852. [Online]. Available: https://onlinelibrary.wiley.com/doi/abs/10.1002/9781118511305.ch23
- [48] I. F. Akyildiz, D. Pompili, and T. Melodia, "Underwater acoustic sensor networks: Research challenges," *Ad Hoc Netw.*, vol. 3, no. 3, pp. 257–279, May 2005. [Online]. Available: https://www.sciencedirect. com/science/article/pii/S1570870505000168
- [49] K. Casey, A. Lim, and G. Dozier, "A sensor network architecture for tsunami detection and response," *Int. J. Distrib. Sensor Netw.*, vol. 4, no. 1, pp. 27–42, Jan. 2008, doi: 10.1080/15501320701774675.
- [50] C.-C. Kao, Y.-S. Lin, G.-D. Wu, and C.-J. Huang, "A comprehensive study on the Internet of Underwater Things: Applications, challenges, and channel models," *Sensors*, vol. 17, no. 7, p. 1477, Jun. 2017. [Online]. Available: https://www.sciencedirect.com/science/ article/pii/S1570870505000168
- [51] E. Felemban, F. K. Shaikh, U. M. Qureshi, A. A. Sheikh, and S. B. Qaisar, "Underwater sensor network applications: A comprehensive survey," *Int. J. Distrib. Sensor Netw.*, vol. 11, no. 11, Nov. 2015, Art. no. 896832, doi: 10.1155/2015/896832.
- [52] T. Kanazawa, "Japan trench earthquake and tsunami monitoring network of cable-linked 150 ocean bottom observatories and its impact to Earth disaster science," in *Proc. IEEE Int. Underwater Technol. Symp. (UT)*, Mar. 2013, pp. 1–5.
 [53] S. Jiang, "On reliable data transfer in underwater acoustic networks:
- [53] S. Jiang, "On reliable data transfer in underwater acoustic networks: A survey from networking perspective," *IEEE Commun. Surveys Tuts.*, vol. 20, no. 2, pp. 1036–1055, 2nd Quart., 2018.
- [54] National Weather Service. National Weather Service—Tsunami Hazards. Accessed: Jun. 5, 2021. [Online]. Available: https://www.tsunami.gov/
- [55] B. Benson, "Design of a low-cost underwater acoustic modem for short- range sensor networks," Ph.D. dissertation, Dept. Comput. Sci. Eng., UC San Diego, La Jolla, CA, USA, 2010. [Online]. Available: https://escholarship.org/uc/item/8j70s26c
- [56] GIS Mapping Software, Location Intelligence & Spatial Analytics | ESRI. Accessed: Jul. 3, 2023. [Online]. Available: https://www.esri.com/enus/home
- [57] P. Jiang, X. Wang, and J. Liu, "A sensor redeployment algorithm based on virtual forces for underwater sensor networks," *Chin. J. Electron.*, vol. 27, no. 2, pp. 413–421, Mar. 2018. [Online]. Available: https://ietresearch.onlinelibrary.wiley.com/doi/abs/10.1049/cje.2017.10. 003
- [58] X. Pan, J. Zhu, M. Liu, X. Wang, Z. Peng, J. Liu, and J. Cui, "An on-demand scheduling-based MAC protocol for UW-WiFi networks," *J. Mar. Sci. Eng.*, vol. 11, no. 4, p. 765, Mar. 2023. [Online]. Available: https://www.mdpi.com/2077-1312/11/4/765
- [59] A. Almutairi and S. Mahfoudh, "Deployment protocol for underwater wireless sensors network based on virtual force," *Int. J. Adv. Comput. Sci. Appl.*, vol. 8, no. 11, pp. 241–249, 2017.
- [60] D. N. Sandeep and V. Kumar, "Review on clustering, coverage and connectivity in underwater wireless sensor networks: A communication techniques perspective," *IEEE Access*, vol. 5, pp. 11176–11199, 2017.
- [61] S. Li, W. Wang, and J. Zhang, "Efficient deployment surface area for underwater wireless sensor networks," in *Proc. IEEE 2nd Int. Conf. Cloud Comput. Intell. Syst.*, vol. 3, Oct. 2012, pp. 1083–1086.

- [63] L. Liu, M. Ma, C. Liu, and Y. Shu, "Optimal relay node placement and flow allocation in underwater acoustic sensor networks," *IEEE Trans. Commun.*, vol. 65, no. 5, pp. 2141–2152, May 2017.
- [64] B. Kahriman, A. M. Demirtas, and B. Tavli, "Surface gateway placement optimization for underwater sensor networks," in *Proc. 28th Telecommun. Forum (TELFOR)*, Nov. 2020, pp. 1–4.
- [65] T. Ayub, S. Deb, and S. M. Shaby, "Surface-level gateway deployment for underwater sensor networks," in *Proc. Int. Conf. Innov. Inf., Embedded Commun. Syst. (ICIIECS)*, Mar. 2015, pp. 1–6.
- [66] H. Albarakati, R. Ammar, R. Elfouly, and S. Rajasekaran, "Geometryassisted multi surface-gateways placement topologies for underwater sensor networks," in *Proc. IEEE Int. Symp. Signal Process. Inf. Technol.* (ISSPIT), Dec. 2019, pp. 1–7.
- [67] J. Huang, L. Sun, X. Wei, P. Sun, H. Huang, and R. Wang, "Redundancy model and boundary effects based coverage-enhancing algorithm for 3D underwater sensor networks," *Int. J. Distrib. Sensor Netw.*, vol. 10, no. 4, Apr. 2014, Art. no. 589692, doi: 10.1155/2014/589692.
- [68] M. C. Domingo, "Optimal placement of wireless nodes in underwater wireless sensor networks with shadow zones," in *Proc. 2nd IFIP Wireless Days (WD)*, Dec. 2009, pp. 1–6.
- [69] N. Bahrami, N. H. H. Khamis, and A. B. Baharom, "Study of underwater channel estimation based on different node placement in shallow water," *IEEE Sensors J.*, vol. 16, no. 4, pp. 1095–1102, Feb. 2016.
- [70] C. Venkateswara Rao, M. R. Sankar, T. N. Prasad, R. Devi, M. S. U. Suseela, V. Praveena, and Y. Srinivas, "Comparison of acoustic channel characteristics in shallow and deep-sea water," in *Cognitive Computing and Cyber Physical Systems*, N. Gupta, P. Pareek, and M. Reis, Eds. Cham, Switzerland: Springer, 2023, pp. 256–266.
- [71] (2016). Offshore Oil Production in Deepwater and Ultra-deepwater is Increasing. Accessed: May 28, 2023. [Online]. Available: https:// www.eia.gov/todayinenergy/detail.php?id=28552
- [72] (Jul. 2021). NCEI/WDS Global Historical Tsunami Database, 2100 BC to Present. Accessed: May 28, 2023. [Online]. Available: https:// www.ncei.noaa.gov/access/metadata/landing-page/bin/iso?id=gov.noaa. ngdc.mgg.hazards:G02151
- [73] Marine Geoscience Data System. *GeoMapApp*. Accessed: Jul. 3, 2023. [Online]. Available: http://www.geomapapp.org/
- [74] M. Y. I. Zia, J. Poncela, and P. Otero, "State-of-the-art underwater acoustic communication modems: Classifications, analyses and design challenges," *Wireless Pers. Commun.*, vol. 116, no. 2, pp. 1325–1360, Jan. 2021, doi: 10.1007/s11277-020-07431-x.
- [75] P. Xie, Z. Zhou, Z. Peng, H. Yan, T. Hu, J.-H. Cui, Z. Shi, Y. Fei, and S. Zhou, "Aqua-sim: An NS-2 based simulator for underwater sensor networks," in *Proc. OCEANS*, Oct. 2009, pp. 1–7.
- [76] H. Luo, K. Wu, R. Ruby, F. Hong, Z. Guo, and L. M. Ni, "Simulation and experimentation platforms for underwater acoustic sensor networks: Advancements and challenges," *ACM Comput. Surv.*, vol. 50, no. 2, pp. 1–44, May 2017, doi: 10.1145/3040990.
- [77] S. Sendra, J. Lloret, J. M. Jimenez, and L. Parra, "Underwater acoustic modems," *IEEE Sensors J.*, vol. 16, no. 11, pp. 4063–4071, Jun. 2016.



AHMED ALJUGHAIMAN (Member, IEEE) received the B.S. degree in computer and information technology (computer networks and information security) from Indiana University— Purdue University Indianapolis, in 2011, the master's degree in network security from DePaul University, in 2013, and the master's degree in information assurance and the Ph.D. degree in security engineering from the University of Colorado Colorado Springs, in 2019 and 2021,

respectively. He is currently an Assistant Professor with the College of Computer Sciences and Information Technology, King Faisal University, Saudi Arabia. His current research interests include underwater wireless sensor networks, underwater communications, software-defined networks, cybersecurity, computer networks, terrestrial wireless sensor networks, network protocols, network security, the Internet of Things, blockchain, and unmanned aerial vehicles.